Author's response to reviews

Title: Safety of liver resection and effect on quality of life in patients with benign hepatic disease: Single center experience

Authors:

Carsten Kamphues (carsten.kamphues@charite.de)
Sabine Engel (carsten.kamphues@charite.de)
Timm Denecke (carsten.kamphues@charite.de)
Roberta Bova (carsten.kamphues@charite.de)
Michael Hippler-Benscheidt (carsten.kamphues@charite.de)
Gero Puhl (carsten.kamphues@charite.de)
Peter Neuhaus (carsten.kamphues@charite.de)
Daniel Seehofer (carsten.kamphues@charite.de)

Version: 4 Date: 13 July 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

I am sending you my revised manuscript with the following changes for publication in BMC Surgery:

Major revisions (we require the author to make these changes)

References: Please make sure references are cited in numerical order. Currently you cite reference 19 before reference 18.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Authors Contributions: this should be formatted as a paragraph, rather than a list.

Tables: please remove the vertical lines from your tables

All revisions were done according to your suggestions.

The manuscript has been checked again and I do not have any further corrections.

Sincerely,

Carsten Kamphues